Author's response to reviews

Title: The prevalence and presentation pattern of developmental dental hard-tissue anomalies in children from a suburban population in Nigeria

Authors:

Dada O Temilola (tesy4jil2004@yahoo.com)
Morenike O Folayan (toyinukpong@yahoo.co.uk)
Olawumi A Fatusi (ofatusi@yahoo.com)
Nneka M Chukwumah (nnechukwumah@yahoo.com)
Nneka Onyejaka (nnekaonyejaka@yahoo.com)
Elizabeth O Oziegbe (elioziegbe@yahoo.com)
Titus A Oyedele (ayotitus4christ@gmail.com)
Kikelomo A Kolawole (kikelomokolawole@gmail.com)
Hakeem Agbaje (kimexy@gmail.com)

Version: 5 Date: 17 July 2014

Author's response to reviews: see over
Dear Editor,

**Cover Letter**

We the authors of this manuscript titled ‘The prevalence of dental anomalies in children from a suburban Nigerian population’ would like to inform you of the changes made to this manuscript.

We have made the following changes to the manuscript:

- The authors have declared that there is NO competing interest in the manuscript.
- The authors have clearly stated all the contributions of each author of this manuscript.
- The authors have also done the line numbering of the manuscript.
- The authors have also acknowledged individual that should be acknowledged due to his contributions to this manuscript.
- The authors have also placed all the tables in this manuscript after the references.

We believe we have done all the required revision for this manuscript.

Thanks

Olawaseyide Temilola
Obafemi Awolowo University Teaching Hospitals Complex, Ile-Ife
E-mail: tesy4jl2004@yahoo.com
Tel: +2348167258512